Author's response to reviews

Title: The course of Depression in Late Life as measured by the Montgomery and Aasberg Depression Rating Scale in an observational study of hospitalized patients.

Authors:

- Tom Borza (tom.borza@sykehuset-innlandet.no)
- Knut Engedal (Knut.engedal@aldringoghelse.no)
- Sverre Bergh (sverre.bergh@sykehuset-innlandet.no)
- Jurate S Benth (jurate.saltyte-benth@medisin.uio.no)
- Geir Selbaek (geir.selbaek@aldringoghelse.no)

Version: 2 Date: 3 February 2015

Author's response to reviews:

The cover letter is kept unchanged as the changes made in the current version are mainly placement issues.